Sampling notes, La Perouse cruise, 2018-39, 7 – 18 May 2018
CTD/Rosettes:
96 files, 0001 to 0184.
7 stand-alone CTD casts (one was a test).

59 actual rosette casts, 30 casts with no Niskin (needed).
0 cast needed two files.
911 CTD:  443;   SBE25:  334.

Configuration file:
We started with: 2018-39-Start.xmlcon.
Standard procedure:

The rosette was brought to the surface.  Pumps were turned ON.  Then the rosette was brought down to 10m and back up.  Data started to be archived then, with the rosette at the surface for 30 seconds longer.  Then the cast would start. 

For the SBE25: the fish was turned on, CTD was lowered in the water, brought down to 10m and back up.  The CTD then stayed at the surface until two minutes after the CTD was turned on, then the profile was started. 

Problems with CTD:

The SBE25 has a problem with the pressure sensor.  It was only used for 6 casts: 
Station E1, cast 11

Station C1, cast 16
Station C2, cast 18

Station C3, cast 20
Station B8, cast 24
Station B7, cast 26
Station LBP8, cast 85 (test only)

Cast 0088 – LBP7: the header seems to be missing some information?
Niskins & Samples at CTD/Rosette casts:
Cast 0001 – SI: test cast only, 24 Niskins closed but no file needed.  Nina did take one HPLC sample but does not need data file.
Cast 0037 – LB15: Niskins 11 and 24 closed just to check integrity, not needed. 

Cast 0048 – LC09: that cast was supposed to be done “out of order” but the Niskins got closed in order, so three Niskins were closed at the wrong depth. 

Cast 0091 – LBP5: one Niskin closed at the wrong depth and one Niskin with no sample (spigot open).
Cast 0101 – CPE2: Niskin 11 closed just to check integrity, not needed.

Cast 0114 – CS02: No sample from Niskin 11 – the Niskin broke. 

Cast 0164 – 12: Niskins all closed out of order and at the wrong depth. Check on rosette log to see what happened!

File names – Headers:
File 0157 – 14: named 0158 on the rosette log.
Casts with no PAR sensor:
The PAR sensor was on for all casts done with CTD 0443.
Casts out of order: 
0036 (LB14) – 0037 (LB15, see above) – 0059 (LC04) – 0081 (LG09) – 0086 (LBP8) – 0094 (LBP3) – 0114 (CS02) – 0135 (CS09) – 0154 (JS) – 0164 (12) 
Loops:

See loops log.
6 loops were taken at stations during the 5m Niskin, and 11 underway loops were collected.  All were:  2x Chl, 1x Sal. 
TSG: 
There is no flow meter on this TSG.  The remote temperature is working. 

Two files were created. 
